Author’s response to reviews

Title: A retrospective analysis of breast cancer subtype based on ER/PR and HER2 status in Ghanaian patients at the Korle Bu Teaching Hospital, Ghana

Authors:

Bernard Seshie (seshiegh2000@yahoo.com)
Nii A Adu-Aryee (naduaryee@chs.edu.gh)
Florence Dedey (flodedey@yahoo.com)
Benedict N.L Calys-Tagoe (calys75@hotmail.com)
Joe N Clegg-Lamptey (clegglamptey@chs.edu.gh)

Version: 2 Date: 10 September 2014

Author’s response to reviews:

Dear Sir,

A retrospective analysis of breast cancer subtype based on ER/PR and HER2 status in Ghanaian patients at the Korle Bu Teaching Hospital, Ghana:

Revision for Page Numbering

I have made the following changes:

I have continued page numbering for the whole main text, including References

Joe Nat Clegg-Lamptey
Corresponding author